Author's response to reviews

Title: Impact of ethnic-specific guidelines for anti-hypertensive prescribing in primary care in England: a longitudinal study

Authors:

Lena Barrera (l.barrera-vergara08@imperial.ac.uk)
Craig Leaper (craig.leaper@gmail.com)
Utz J. Pape (u.pape@imperial.ac.uk)
Azeem Majeed (a.majeed@imperial.ac.uk)
Marta Blangiardo (m.blangiardo@imperial.ac.uk)
Christopher Millett (c.millett@imperial.ac.uk)

Version: 3
Date: 14 February 2014

Author's response to reviews: see over
London 14 of February 2014

Mr
David Sclar
Journal Editorial Office
BioMed Central

Refer to 2119822237103735 "Impact of ethnic-specific guidelines for antihypertensive prescribing in primary care in England: a longitudinal study"

Dear David

Thank you for accepting our article for publication. We have made the required amendments in the main text of the article. The competing interest, the author's contribution and the acknowledgement sections, and the tables were moved regarding the BMC editorial instructions. The box was formatted as a table.

Thank you again for approving our article and we hope to have a new opportunity to report our research in your journal.

Kings regards,

Lena Barrera
On behalf of authors